Check for updates

OPEN ACCESS

EDITED BY Marc Jean Struelens, Université libre de Bruxelles, Belgium

REVIEWED BY

Carlo Biz, University of Padua, Italy Eric Toussirot, INSERM CIC1431 Centre d'Investigation Clinique Besançon, France

*CORRESPONDENCE Shiloh Plaut ⊠ shilowhale@gmail.com

SPECIALTY SECTION

This article was submitted to Infectious Diseases: Pathogenesis and Therapy, a section of the journal Frontiers in Medicine

RECEIVED 24 May 2022 ACCEPTED 27 February 2023 PUBLISHED 06 April 2023

CITATION

Plaut S (2023) "Long COVID-19" and viral "fibromyalgia-ness": Suggesting a mechanistic role for fascial myofibroblasts (Nineveh, the shadow is in the fascia). *Front. Med.* 10:952278. doi: 10.3389/fmed.2023.952278

COPYRIGHT

© 2023 Plaut. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

"Long COVID-19" and viral "fibromyalgia-ness": Suggesting a mechanistic role for fascial myofibroblasts (Nineveh, the shadow is in the fascia)

Shiloh Plaut[®]*

Department of Basic and Clinical Sciences, University of Nicosia Medical School, Nicosia, Cyprus

The coronavirus pandemic has led to a wave of chronic disease cases; "Long COVID-19" is recognized as a new medical entity and resembles "fibromyalgia" which, likewise, lacks a clear mechanism. Observational studies indicate that up to 30%-40% of convalescent COVID-19 patients develop chronic widespread pain and fatigue and fulfill the 2016 diagnostic criteria for "fibromyalgia." A recent study suggested a theoretical neuro-biomechanical model (coined "Fascial Armoring") to help explain the pathogenesis and cellular pathway of fibromyalgia, pointing toward mechanical abnormalities in connective tissue and fascia, driven by contractile myo/fibroblasts and altered extracellular matrix remodeling with downstream corresponding neurophysiological aberrations. This may help explain several of fibromyalgia's manifestations such as pain, distribution of pain, trigger points/tender spots, hyperalgesia, chronic fatigue, cardiovascular abnormalities, metabolic abnormalities, autonomic abnormalities, small fiber neuropathy, various psychosomatic symptoms, lack of obvious inflammation, and silent imaging investigations. Pro-inflammatory and pro-fibrotic pathways provide input into this mechanism via stimulation of proto/myofibroblasts. In this hypothesis and theory paper the theoretical model of Fascial Armoring is presented to help explain the pathogenesis and manifestations of "long COVID-19" as a disease of immuno-rheumo-psycho-neurology. The model is also used to make testable experimental predictions on investigations and predict risk and relieving factors.

KEYWORDS

SARS CoV-2, COVID-19, pain, myofibroblast, fascial armoring, fibromyalgia, pathogenesis, psychosomatic

1. Introduction

The coronavirus pandemic has impacted our world not only because of the acute phase and its mortality. "Long/Post COVID-19" is recognized as a new medical entity and resembles functional psychosomatic syndromes (1–4). The post-acute sequelae of COVID-19 is being diagnosed not only in those who developed severe acute COVID-19, but also in infected individuals who had mild and even asymptomatic cases (3). In current literature there is no consistent term for the post-acute sequalae of SARS-CoV-2 infection. In this paper, the term "long COVID-19" will be used.

Recent investigations indicate that most "long COVID-19" patients present with lasting symptoms that impact morbidity and quality of life, ranging from neuropsychiatric,

cardiovascular, pulmonary, gastrointestinal, musculoskeletal, reproductive, immunological, otolaryngologic, and dermatologic symptoms, up to multiorgan systemic dysfunction (2). Studies have been reporting various incidence rates for "long COVID-19." According to some estimates, long-lasting symptoms following COVID-19 affect as much as 76% of people at 6 months post-infection (1). The United Kingdom office for national statistics estimated that the five-week prevalence of any symptom was 22.1%, and the 12-week prevalence was 9.9% (1).

The puzzle of the similarity between "long COVID-19" and functional somatic syndromes is intriguing to many (1, 3, 4). A recent study (4) suggested a theoretical model (coined "Fascial Armoring") with a cellular pathway to help explain part of the mechanism and pathogenesis of fibromyalgia-like and "functional psychosomatic syndromes." Psychosomatic syndromes are suggested to be on a spectrum of a single medical entity (5), which may share a common rheuma-psycho-neurological mechanism (4). Fascial armoring describes the rheumatological facet of this medical entity. It suggests that these syndromes are driven, in part, by myofibroblast-generated-tensegrity-tension in the fasciomusculoskeletal system. This work offers a theoretical model with a cellular pathway to help explain the mechanism of "long COVID-19." In light of the findings, the discussion proposes that fascia may have a role in the mechanism of "long COVID-19," which may involve myofibroblast-mediated-biotensegrity-tension.

The paper is organized as follows: section 2 presents the hypothesis of this work. Section 3 presents the theoretical model of fascial armoring. Section 4 gives an overview of "long COVID-19" manifestations and SARS-CoV-2-related fibrosis, and a suggested mechanistic link between SARS-CoV-2 infection and fibromyalgia-like syndromes. Then, the discussion in section 5 provides an explanation of "long COVID-19" symptomatology in light of this theoretical model. Afterward, predictions of expected results of future investigations of "long COVID-19" and risk and relieving factors are presented based on the suggested model. Finally, section 6 is a summary and conclusion.

2. Hypothesis

The central hypothesis presented in this work is that SARS-CoV-2 infection or acute COVID-19 can result in a myofascial and connectivetissue fibrotic disease by providing input (via immune signals and inflammatory mediators) into the positive feedback cascade of fascial armoring. This is hypothesized to lead to alterations in fascia's remodeling mechanisms and a widespread induction of proto/ myofibroblasts in myofascial and/or soft tissue extracellular matrix (ECM), mediating the transformation of fascial tissue. The cellular pathway of the pathophysiology of fibromyalgia was recently suggested to involve mechanosensitive signaling of myofibroblasts in soft tissue (e.g., integrin, focal adhesion, actin, Rho-associated-kinase, and transforming growth factor beta pathways) (4). Changes in the ECM have important effects on cells and physiological processes (6). Chronic tension in myofascial tissue and fascial dysfunction are expected to be hallmarks of fibromyalgia-like diseases (4). Therefore, due to its similarity with other functional somatic syndromes, "long COVID-19" is speculated to involve myofibroblast-mediated-biotensegrity-tension and mechanical compression in the fasciomusculoskeletal system along with immune-related processes (e.g., viral persistence, immune dysregulation, subclinical neuroinflammation, etc.). In severe cases, it would manifest as a global chronic exertional compartment-like syndrome (4, 7) impeding the function of various organ systems and hallmarked by multiorgan insidious progression as well as malaise and exertional fatigue. Figure 1 depicts the general overlap of conditions according to this framework where variation in manifestation depends, in part, on the overlap of anatomical areas of involved fascia and the severity of fascial armoring (4).

3. The theoretical model of "fascial armoring"

The "fascial armoring" model describes how contractile cells in myofascial tissue can help explain the symptomatology of



FIGURE 1

Overlaps of fibromvalgia-related and myofascial pain syndromes [adapted for long COVID-19 from Figure 7 in Plaut (2022) (4)]. Not all relationships and overlaps are depicted in this scheme. The colors have no special meaning. Variation in manifestation depends, in part, on the anatomical area and severity of involvement of the biotensegrity structure and the downstream consequences of myofibroblast-generated-tensegrity-tension and fascial stiffness. If local fascial armoring becomes widespread, it will manifest as a fibromyalgia-like condition, depending on anatomical location, severity, and the layers involved, as well as additional soft tissue and visceral organ involvement. Psychological, metabolic, immune endocrine, autonomic, behavioral, and other factors also contribute to variation in manifestation. The term "Healthy" is open to interpretation. "Nonspecific low back pain" reflects fascial armoring and imbalanced tensegrity forces in the lower back. "Nonspecific neck pain" reflects imbalance affecting the neck. "Primary headache disorders" i.e., imbalance in the head. "Irritable bowel syndrome" reflects fascial armoring and compressive forces in the lower back and abdomen (which might affect peristalsis and gut microbiome bidirectionally). "Myofascial pelvic pain syndrome" (including "interstitial cystitis")-fascial armoring and fascial stiffness/pressure in the pelvic fascia, etc. If fascial armoring becomes more widespread, it can be expressed as somatic symptom disorder or fibromyalgialike conditions. Long COVID-19 is included here in cases when this post infectious syndrome manifests as a fibromyalgia-like syndrome with no clear pathology found, normal investigations and imaging and laboratory tests, and with marked discrepancy between patient complaints and clinical findings. CWP, chronic widespread pain; ME/ CFS, myalgic encephalomyelitis/chronic fatigue syndrome; MPS, myofascial pain syndrome; SSD, somatic symptom disorder Multiplying entities when there really is only one is not beneficial for science

fibromyalgia-like syndromes. In this section, an explanation of the model of "fascial armoring" is provided, as well as a description of cellular pathways linking COVID-19 to a myofascial bio-tensegrity abnormality. Empirical findings will be presented for explaining the model of fascial armoring with the purpose of suggesting a role for fascial myo/fibroblasts in "long COVID-19."

3.1. A brief explanation of "fascial armoring" and its cellular pathway

The following findings are the basis for the theoretical model that is proposed in this work:

- (i) Fascia is a continuous three-dimensional network of connective tissue that permeates the body at various layers and depths, connects and envelops organs and muscles, while undergoing constant remodeling and exhibiting qualities of tensegrity (or bio-tensegrity) (4, 6, 8, 9). According to the principle of tensegrity, structures are stabilized under conditions of continuous tension with discontinuous compression and behave as one connected structure (6, 10). "Bio-tensegrity" applies the principles of tensegrity to our understanding of the dynamics of living organisms (11). Fascia transmits forces to a distance through mechanical links and myofascial chains (9). In humans, most skeletal muscles are directly linked by connective tissue (9). Myofascial tissue is even continuous with the meninges (12).
- (ii) Fibroblasts are a main cell type synthesizing and regulating the ECM. When subjected to mechanical tension (and other signaling cues), fibroblasts differentiate into protomyofibroblasts, a phenotype that contains cytoplasmic actin stress fibers terminating in fibronexus adhesion complexes (13, 14). These adhesion complexes serve to bridge the internal cytoskeleton of the proto/myofibroblast and its integrins with the surrounding ECM. The structural linkage created allows proto/myofibroblasts to generate traction and transfer contractile forces to the nearby ECM. Forces in the ECM are maintained and reinforced over time by matrix remodeling and further deposition of collagen and ECM material (13). The myofibroblast phenotype has increased expression of alphasmooth muscle actin (α -SMA) fibers that enable them to maintain mechanical tension in the surrounding ECM in a mechanism known as slip and ratchet (13). A positive feedback regulation occurs with α -SMA and force generation, facilitated by transforming growth factor beta 1 (TGF- β 1), and it is a basis for a vicious cycle that propels myofibroblast activity in granulation tissue, so cells can generate considerable amount of force. Myofibroblasts have similar cellular behavior independent of their anatomical location (15) although "fibroblasts" are a diverse family of cells (16, 17). It is possible that the protein conformation of α-SMA in myofibroblasts of fibrotic muscles differs from that of myofibroblasts of fibrotic lungs or in smooth muscle cells. Not all myofibroblasts express α -SMA (13) and using it as a marker for fibrogenic cell activity in skeletal muscle may be problematic (18).
- (iii) Myofibroblasts are found in the fascia and interstitial ECM of healthy individuals. They have a role in maintaining basal

mechanical tissue tone (19, 20) and function as a large network (21). Soft tissue fibroblasts form a widespread interconnected cellular network with potentially critically overlooked physiological and functional importance (22).

(iv) Fascia contains a dense network of sensory nerve endings that play a part in pain perception (23, 24).

The myofascial continuum in a tensegrity framework is the basis for the model of fascial armoring, which is portrayed as a global chronic compartment-like syndrome in severe cases. Empirical evidence that may support this fascia-based pathogenesis of fibromyalgia can be found in studies such as:

- 1) A study measured significantly higher intramuscular pressures in the trapezius muscle of fibromyalgia patients with a mean value of 33.48±5.90 mmHg while the mean pressure measured in rheumatic disease controls was 12.23±3.75 mmHg (25). The authors of that study note that the burden of the pressure abnormality in fibromyalgia might contribute to diffuse muscle pain and may be an intrinsic feature of the disease. The mean blood pressure is approximately 30 mmHg at the end of arterioles (26). Maintaining normal pressure gradients and starling forces is critical for proper perfusion in tissues and normal health. Chronic exertional compartment syndrome can be diagnosed when at least one of the following intramuscular compartment pressure measurements is measured: (i) pre-exercise pressure≥15mmHg (ii) 1-min post-exercise pressure of ≥30 mmHg, or (iii) 5-min post-exercise pressure≥20 mmHg (27). Fibromyalgia as a disease of exclusively central neuroplastic pain should be reconsidered (25).
- Fibromyalgia patients have decreased peripheral blood flow, which suggests there are functional disturbances in the cardiovascular system (28). There is evidence indicating a decreased transcapillary permeability in fibromyalgia patients (29) and abnormalities in arteries (30).
- Biopsies of fibromyalgia cases indicate definite and nonspecific muscle changes which are suggested to result from chronic muscle contraction and ischemia of unknown etiology (31).
- 4) A study measuring muscle damping in fibromyalgia patients revealed elevated levels in all 14 sampled patients. This finding reflects increased muscle tension. The mean value in the fibromyalgia patient group was more than twice that of controls and maximum values were more than three-fold higher (32).
- 5) In muscles of fibromyalgia patients, increased deoxyribonucleic acid fragmentation and ultrastructural changes are suggested to be secondary to chronic muscle contraction (33).
- A biopsy study of fibromyalgia patients found that peripheral fibroblast TGF-β gene expression is significantly higher compared to controls (34).
- 7) A study investigating inflammatory factors in skin biopsies from fibromyalgia patients showed elevated levels of interleukin 1 β (IL-1 β), in 38% of patients, IL-6 in 27%, and tumor necrosis factor-alpha (TNF- α) in 32% of patients. None of the cytokines could be detected in skin samples of healthy controls (35). These factors, along with TGF- β , are among the main mediators of myofibroblast activity (14, 36). Fibromyalgia patients have a unique cytokine profile of IL-6, IL-8, IL-10, TNF- α , and eotaxin/CCL24 (37).

- 8) The protein expression of genes involved in ECM turnover and oxidative metabolism has a differential expression in fibroblast of fibromyalgia patients, which could explain the inflammatory status of these patients (38).
- 9) Interstitial concentrations of metabolic substances in fibromyalgia were measured in a microdialysis study (39). Lactate, glutamate, and pyruvate concentrations were found to be significantly higher in fibromyalgia patients. Increased dialysate lactate in response to acetylcholine stimulation was shown in fibromyalgia (40).
- 10) Abnormalities in collagen crosslinking may contribute to remodeling of the ECM and deposition of collagen around the nerve fibers, which might lead to a lower pain threshold in fibromyalgia (41). A significantly lower amount of intramuscular collagen was found in fibromyalgia patients, which may cause a lower threshold for muscle micro-injury and result in non-specific signs of muscle pathology (42).

Even if myofibroblasts eventually de-differentiate to fibroblasts or undergo apoptosis, they would leave behind a remodeled dysfunctional fascia with altered biomechanical properties of the ECM. Tension and mechanical compression would lead to the activation of mechanoreceptors and nociceptors in nerve endings embedded in fascia, driving pain signals to the spinal cord and brain (4). Reducing the muscle pressure to normal levels in fibromyalgia may therapeutically alter the clinical picture (25). ECM forces causing stretch lesions in peripheral nervous tissue might affect sensory and sympathetic fibers, potentially causing further pain, radiation of pain, and autonomic abnormalities (4). The deep fascia is innervated with sympathetic nerve fiber endings (43), which suggests that the emotional and mechanical aspects of the disease are deeply integrated in the bio-tensegrity system.

Theoretically, chronic mechanical forces reaching the dura or intracranial/spinal fascia via myofascial chains and tensegrity dynamics might have significant physiological effects on the central nervous system. For example, mechanical stress increases amyloid beta, tau, and alpha-synuclein in the brain of wild type mice (44). Systemic circulating inflammatory mediators are expected to affect intracranial ECM and the brain. The downstream effects of chronic nociception due to fascial armoring would be similar to the effects of pain arising from other chronic painful conditions (e.g., increased neuroendocrine and hypothalamic-pituitary-adrenal axis activity, cortico-limbic and dopaminergic response to pain, brain neuroplasticity, sympathetic activity, microglia activation and possible excitotoxicity, central neuroinflammation, peripheral neurogenic inflammation, and so forth). While this paper did not focus on the central nervous system, it is noted that neuroendocrine and humoral pathways are part of fascial armoring and may help explain this complex entity. A more detailed mechanistic explanation of the above phenomena was provided in a recent review (4).

3.2. Elaboration on the molecular pathway linking inflammation to "fascial armoring" and induction of myofibroblasts

Upon activation, TGF- β 1 promotes the formation and activity of myofibroblasts (45). Most of the TGF- β molecules are not

synthesized on demand but are present in the ECM in latent form and can be activated when required (36). TGF-B1 can induce fibrosis by activating both canonical-small mother against decapentaplegic (SMAD) based and non-canonical (non-SMADbased) signaling pathways. This, in turn, results in activation of myofibroblasts and superfluous production of ECM, while at the same time inhibiting ECM degradation (46). Non-SMAD pathways such as mitogen-activated protein kinase (MAPK), extracellularsignal-regulated kinase (ERK), JUN N-terminal kinase (JNK), RHO-associated kinase (ROCK), p38, and AKT pathways can also be activated by TGF-B1 (46, 47). SMAD2/3-SMAD4 is a heterodimer that often functions together with an AP-1 complex to mediate TGF-\beta-induced transcription. Upon cell damage and release of inflammatory substances from macrophages and T-cells, resident fibroblasts become activated by signals such as IL-6, TNF- α , and TGF- β . The activated fibroblast assumes a spindle shape and expresses high-affinity type 2 TGF-ß receptor and begins to produce more collagen (47). Platelet derived growth factor (PDGF) has four isoforms that bind two PDGF receptor tyrosine kinases (PDGFR α and β). Fibroblasts and myofibroblasts express PDGFRs which can promote survival, proliferation, and migration when they bind their ligand PDGF (47). Mast cell degranulation of tryptase can induce myofibroblast differentiation and collagen synthesis directly and independently of TGF- β (48). Importantly, IL-1, IL-6, IL-11, IL-17, and IL-25, are cytokines that induce and enhance myofibroblasts (49). Along with integrins that can directly activate latent TGF-β induced pro-fibrotic pathways, inflammatory cytokines including IL-6 can have profound effects on the pro-fibrotic activity of fibroblasts (49).

Myofibroblasts fate is determined by complex mechanisms of cellular signaling. They can revert back to the fibroblast phenotype, or they can undergo apoptosis and self-clearance. Evasion from these mechanisms can lead to pathological and sustained fibrotic responses (47). Rho induces the nuclear translocation of Yes-associated protein/TAZ which promotes the expression of genes important for ECM, such as PAI-1, connective tissue growth factor (CTGF), and COL1. Stiffening of connective tissue directly drives fibrosis by controlling the integrin-mediated activation of latent TGF-\$1. The mechanosensitive factor YAP-1 and its downstream signaling pathways of integrin beta-1 promote fibrosis (50). CTGF/ CCN2 expression is induced by several stimuli including growth factors, cytokines, oxygen deprivation, mechanical stress, and more. CTGF/CCN2 has many functions (e.g., ECM turnover and cell behavior) and most TGF-B responses ultimately involve its stimulation (51).

Activation of certain toll-like receptors (TLRs) augments myo/ fibroblast differentiation and fibrosis (52). Interleukin-1 receptor (IL-1R) kinase 4 (IRAK-4) has a phosphorylation activity that is essential for IL-1R/TLR-mediated cellular processes. A 2007 study showed the importance of IRAK-4 kinase activity in TLR-mediated myeloid differentiation factor 88-dependent signaling (53). That study investigated the single stranded ribonucleic acid (ssRNA) sensitive pattern recognition receptor TLR7. Also, neutrophil extracellular traps induce the epithelial to mesenchymal transition (54). Macrophages play a part in fibrotic changes in tissue and alter collagen crosslinking and matrix remodeling (*via* matrix metalloproteases, enzymatic secretion, paracellular signaling, macrophage-tomyofibroblast transition, etc.) (55, 56).

4. Long COVID-19

The following findings are presented with the purpose of outlining the manifestation of "long COVID-19" and the mechanism of COVID-19 induced tissue fibrosis.

4.1. Symptoms of "long COVID-19" and complications of SARS-CoV-2 infection

"Long COVID-19" is currently a recognized term in scientific literature. The Centers of Disease Control (CDC) and National Institute of Health (NIH) describe "long COVID-19" patients as individuals with lasting symptoms of COVID-19 that persist for more than 4 weeks since initial infection (1, 57). Manifestations of "long COVID-19" include:

4.1.1. Fatigue and exertional fatigue

Unrelenting exhaustion, often reported as a constant state of weariness that reduces a person's energy, motivation, and concentration, is common after COVID-19 (1) and may last more than 12 months after initial infection (58) often occurring independent of the severity of acute infection (58). "Long COVID-19" fatigue has much overlap with myalgic encephalomyelitis/chronic fatigue syndrome (ME/CFS) and many patients with "long COVID-19" fulfill the diagnostic criteria for ME/CFS (3). In a cross-sectional study, only 4% of COVID-19 patients had persistently abnormal findings on chest X-ray after recovery, while above 60 percent of patients did not feel back to full health more than 6 weeks after recovery from the acute phase (58).

4.1.2. Dyspnea

Some patients with COVID-19 develop a fibrotic state in the lungs with ongoing dyspnea (1, 59). A study assessed the sequelae of COVID-19 at nearly 1 year following diagnosis while focusing on convalescent patients who had non-severe COVID-19 (60). Shortness of breath, fatigue, and sleep difficulties were among the more frequent symptoms. Pulmonary diffusion impairment was seen in 24% of non-severe cases. 56.7% of patients had abnormal CT findings, including ground-glass opacities, fibrosis, bronchiectasis, lines and bands, and nodules (60). Subjective respiratory symptoms are commonly reported after acute COVID-19 disease, but do not correlate with the severity of COVID-19 nor with pulmonary function (61).

4.1.3. Cardiovascular abnormalities

A number of abnormalities can occur post COVID-19 (1): (i) Chronic inflammation of cardiomyocytes can cause myositis and death of cardiomyocytes; (ii) Autonomic abnormalities such as postural orthostatic tachycardia syndrome may arise due to dysfunction of the afferent autonomic nervous system; (iii) Fibroblasts in heart secrete ECM proteins when exposed to prolonged inflammation and cellular damage, resulting in fibrosis.

4.1.4. Cognitive impairment and mental health

Cognitive impairment and psychiatric disorders are reported post COVID-19 (1). More than one third of patients who had non-severe COVID-19 reported anxiety and depression at approximately 1-year follow-up (60).

4.1.5. More symptoms of "long COVID-19 syndrome"

More symptoms of "long COVID-19 syndrome" are cough, chest pain or chest tightness, myalgias, headache, sicca syndrome, rhinitis, poor appetite, diarrhea, dizziness (from orthostasis, postural tachycardia, or vertigo), alopecia, sweating, insomnia, sleep apnea, night sweats, restless legs, vivid dreams, nightmares, and lucid dreams (62-65). Olfactory and gustatory dysfunction are common in the chronic phase (1). Among the most frequent symptoms in COVID-19 survivors at the nearly 1-year follow-up, as found in a Chinese study investigating discharged patients hospitalized during the pandemic, are sleep difficulties (43.3%), shortness of breath (40.8%), fatigue (35.8%), and joint pain (32.5%) (60). Decreased quality of life in 44%-69% of adults was reported when measured 14-21 days after initial COVID test (64). In another study (65) conducted by an online survey across 56 countries, approximately 85% of participants experienced relapses primarily triggered by exercise, physical or mental activity, and stress. Participants experienced an average of 55.9 symptoms, across an average of 9.1 organ systems.

4.2. SARS-CoV-2 infection of various tissues and its related fibrosis

SARS-CoV-2 infection, similar to other infections, can lead to pro-inflammatory and pro-fibrotic processes with scarring and fibrosis of tissues (66). Various organs and tissues are susceptible due to the diverse viral tropism of SARS-CoV-2 (3, 67, 68). Myo/ fibroblast induced fibrotic changes are seen in lungs, heart, liver, kidney, intestines, and several other organs following COVID-19 (48, 60, 67–75). Histopathological abnormalities after coronavirus infection are seen in integumentary, urinary, gastrointestinal, reproductive system, and more (69). Esposito et al. suggest that SARS-CoV-2 can directly invade the enteric and/or parasympathetic nervous system and spread along the vagus nerve or its afferents and innervations of visceral organs (76). Angiotensin converting enzyme 2 (ACE2) and TMPRSS2, the key mediators of SARS-CoV-2 infection (77), were identified in extrapulmonary and musculoskeletal tissue such as muscle cells, endothelial cells, smooth muscle cells, pericytes, muscle stem cells (satellite cells), macrophages, adaptive immune cells (B, T, or natural killer cells), myonuclei (muscle fibers), synovium cells, chondrocytes, and bone cells (osteoblasts and osteoclasts) (78-84). These findings suggest that the musculoskeletal system is a potential site of direct SARS-CoV-2 infection. Along with the multiorgan fibrotic impacts of COVID-19, myofascial tissue is also affected, as shown in the following findings:

4.2.1. Adipocytes

ACE2 is expressed in adipose tissue (85) and there is evidence for replication of SARS-CoV-2 in adipocytes (74, 86). Adipose tissue is involved in SARS-CoV-2 infection in patients with older age, obesity, and diabetes (87). Adipocyte to myofibroblast trans-differentiation is induced by SARS-CoV-2 (88).

4.2.2. Fibroblasts and ECM

SARS-CoV-2 mediates a dysregulation of the renin angiotensin aldosterone system and increased angiotensin II levels which might lead to several chronic and acute diseases. In COVID-19, fibroblast activation due to angiotensin II signaling is systemic, with possible involvement of lungs, heart, and kidneys (85). Another possible consequence of COVID-19 is a deregulation of the ECM homeostatic process, which can lead to excessive deposition and accumulation of disorganized ECM within end organs. ECM fibrotic changes alter the structure of the local tissue, disrupt the organ's normal function, and can lead to dysfunction of an organ-system and eventually even organismal death (36, 89). SARS-CoV-2 infection increases mRNA levels of ACE2, TGF- β 1, CTGF, and fibronectin-1, which are drivers of lung fibrosis (90).

4.2.3. Skeletal muscle and myofascial tissue

Abnormalities in myofascial tissue may be part of COVID-19 and "long COVID-19." Musculoskeletal injury and SARS-CoV-2 related viral myositis possibly results from direct viral invasion of myocytes or due to induction of autoimmune pathways (91). Viral invasion or a direct cytotoxic effect of coronavirus viral particles on skeletal muscle cannot be excluded (91). Viral particles were found in diaphragm and skeletal muscle postmortem following severe COVID-19 (92, 93). Systemic inflammation, hypoxemia, muscle disuse and immobilization, and malnutrition can also contribute to musculoskeletal impairment and fatigue following SARS-CoV-2 infection (91, 94). An autopsy series study of individuals with COVID-19 found signs of myositis (95). These inflammatory features were seen more in skeletal muscles than cardiac muscles. Patients with chronic COVID-19 had more pronounced inflammation although, overall, in most skeletal and cardiac muscles viral load was low or undetectable, suggesting inflammation arises due to circulating viral RNA rather than actual myocyte invasion. Therefore, it is suggested SARS-CoV-2 may be associated with a post-infectious immune-mediated myopathy (95). Another autopsy study of patients who died from COVID-19 found muscle and nerve tissue exhibited inflammatory/immune-mediated damage likely related to release of cytokines. Muscle atrophy, necrotizing myopathy, and myositis were found in psoas muscle samples. The same study found no evidence of direct SARS-CoV-2 invasion of these tissues (96). Histological sections show long COVID-19 patients have muscle fiber atrophy, metabolic alterations, and immune cell infiltration (91). Coronavirus causes strain and tension on skeletal muscles thus leads to myofascial trigger points (97).

Early evidence links "long COVID-19" to myofascial tissue and myofascial pain syndrome as a case report documented a patient with chronic symptoms of exertional dyspnea, brain fog, and myalgia post SARS-CoV-2 infection, with no previous medical history of chronic illness (98). He underwent an extensive multisystem workup that revealed normal pulmonary, cardiac, and end organ functions. His disabling fatigue and chronic pain at 6 months post COVID-19 were defined as "long COVID syndrome" relatively quickly, even though he also met the diagnostic criteria for fibromyalgia. Fibromyalgia Impact Questionnaire score was 88 out of 100 (i.e., severe fibromyalgia). Screening for depression was negative. Treatment with wet and dry needling was initiated for his myalgia and had good short-term therapeutic effects. Interestingly, after having multiple repeated sessions of needling in the torso and neck region achieving temporary pain relief and functional improvement, symptoms migrated/extended to the extremities and lower limbs at 6 months since having begun needling trial. Then, at 12 months since having COVID-19, after faithfully dry needling him in 10 different points with a 21-gauge 1 inch needle again repeatedly, his pain went into remission. This case report may indicate that "long COVID syndrome"-related myalgia might be a form of new-onset myofascial pain (98). Literature is still limited as studies of myofascial and subcutaneous tissue in "long COVID-19" are scarce, but it is unlikely that "central sensitization" mercifully withdrew in this case. A successful exorcism of fibromyalgia requires more than just needles. To the best of the author's knowledge, no study was found to directly examine myofibroblasts in fascia following COVID-19.

5. Discussion

Plausible mechanisms of "long COVID-19" as suggested in literature are: (i) persistent reservoirs of SARS-CoV-2 in certain tissues, (ii) ongoing activity of primed immune cells, (iii) autoimmunity triggered by molecular mimicry, (iv) re-activation of neurotrophic pathogens such as herpesviruses under conditions of COVID-19 immune dysregulation, (v) SARS-CoV-2 interactions with host microbiome/virome communities and microbiome/virome dysbiosis, (vi) clotting/coagulation abnormalities, and (vii) dysfunctional brainstem/vagus nerve signaling (3, 99). These mechanisms may facilitate lasting "long COVID-19" symptoms and work in concert.

Findings from literature presented above show that SARS-CoV-2 and COVID-19 can lead to inflammation with systemically circulating inflammatory mediators and growth factors, and eventually to fibrosis in various tissues, including myofascial tissue. While induction of myofibroblasts normally occurs as a part of the inflammatory and healing process, their persistence in fascia, and the long-term consequences of ECM alterations during the acute and chronic phase, in terms of the fascial bio-tensegrity system, may have negative consequences on the organism. "Long COVID-19" and fibromyalgialike syndromes such as ME/CFS share many characteristics. Some suggest they might share the same pathophysiology (100).

Along the discussion, the term "long COVID-19" is typically used by the author to refer to medically unexplained symptoms after COVID-19 (e.g., fatigue, irritability, cognitive and psychiatric symptoms, pain, etc.) that cannot be attributed to other medical conditions or comorbidities of the patient, and with no clear organic finding to explain them (no pulmonary fibrosis, no autoimmunity, no cardiomyopathy, no cachexia, no reactivated herpes virus, no thyroid disorder or other endocrine dysfunction, negative depression screening, normal clinical examinations, mostly normal routine laboratory tests and imaging investigations, etc.) and marked discrepancy between patient complaints and objective findings, usually labeled by the common clinician as "psychosomatic," "catastrophizing," or factitious.

5.1. Explaining symptomatology of "long COVID-19" in light of the model of "fascial armoring"

"Fascial armoring" as an entity resembling a mild global-chroniccompartment-like syndrome might help explain some known "long COVID-19" manifestations. Non-cardiac chest pain and chest

10.3389/fmed.2023.952278

tightness are common features of fibromyalgia and "long COVID-19." Besides lung fibrosis, a fibrotic contractile fascial tissue in the trunk might be able to mildly impair lung function with possible chest tightness and wheezing. If a chronic mechanical compression develops in the cervical fascia or the superior cervical ganglion, it might disrupt salivary ducts/glands, perhaps leading to sicca. Contractile activity of myofibroblasts and subsequent mechanical tension in the temporal fascia could plausibly lead to a tension-like headache. In the abdomen—aberrant peristalsis and functional gut abnormalities. Pelvic fascia—genitourinary symptoms, and so forth. The main motif guiding the clinical interpretation of this entity throughout this discussion is mechanical tension, compression, and fascial stiffness.

5.1.1. Cardiovascular abnormalities

Peripheral vascular alterations are seen in "long COVID-19" (101), including decreased flow mediated dilation and increased arterial stiffness (102, 103). Vascular abnormalities post COVID-19 were suggested to be due to an inflammatory vasculopathy (104). Vascular structural modification following SARS-CoV-2 infection may be due to the process of endothelial to mesenchymal transition in "long COVID-19" (105), which might increase the stiffness and pre-stress of the blood vessel wall. A global chronic exertional compartment-like syndrome would manifest with chronic fatigue, particularly so upon exertion. New onset of hypertension, as seen in "long COVID-19" might stem, in part, from widespread low-grade mechanical compression of blood vessels and increased resistance in skeletal muscle capillaries. Endothelial damage would result in release of tissue factor.

5.1.2. Fasciomusculoskeletal, joint pain, and myalgia

Musculoskeletal pain in "long COVID-19" may be explained by increased intramuscular pressure, increased myofascial tension, and by sub/clinical chronic ischemia and elevated interstitial nociceptive substances (4). Alterations in skeletal muscle of "long COVID-19" are similar to ME/CFS (91). Potential pathological mechanisms leading to myofascial pain in SARS-CoV-2 infection include hypoxic muscle dysfunctions and psychological stress (97). A study suggested chronic proprioceptor activation might help explain pain in an experimental model of chronic fatigue syndrome (106). Morning stiffnessmyofibroblasts contract over several hours to secure pre-stress and tension in ECM (13), therefore, after a period of rest and immobility a sense of stiffness should be more pronounced. Clinical features of fibromyalgia are common among those who had COVID-19 (107). In a study of 100 participants, fibromyalgia-like symptoms were studied in adult patients with "long COVID-19" via online questionnaires (108). 82% of patients were women; nine participants reported previous pain or an inflammatory condition. The most reported painful sites were chest, lower extremities, head/face, and migrating sites. Generalized pain was self-reported by 75% of participants. New onset fibromyalgia was suspected in 39 out of 100 "long COVID-19" patients, as they fulfilled the 2016 diagnostic criteria for "fibromyalgia." Among the participants fulfilling the 2016 criteria, 23 were completely healthy prior to COVID-19. Many of "long COVID-19" patients develop pain, even if initial infection was mild, and there is an urgent need for pain management in "long COVID-19 syndrome" (108). Fascinatingly, "fibromyalgia" is often perceived as factitious/imagined by many clinicians who do not even believe this entity exists (4, 5). Even though the study above may suffer from biases such as recruitment bias and recall bias, perhaps leading to an overestimation of pain burden, it still deserves sincere consideration without dismissing the findings.

5.1.3. Gastroesophageal reflux disease/acid reflux

Higher incidence of Gastroesophageal reflux disease (GERD)/acid reflux is seen following COVID-19 and has been noted as a manifestation of "long COVID-19" (109). Chronic mechanical disruption of gastroesophageal pressures and sphincter relaxations due to connective tissue compression, tension, and stiffening, might lead to GERD or a GERD-like condition.

5.1.4. Coagulation abnormalities

Widespread fascial tensegrity compression would affect blood flow and turbulence in the vascular system as well as proper vasodilatation of vessels. Virchow's triad consists of hypercoagulability, hemodynamic (turbulence/stasis), and endothelial injury/dysfunction. Therefore, coagulation abnormalities can be expected.

5.1.5. Low-grade fever

Viral particle persistence and immune dysregulation are reasonable explanations for low-grade fever in "long COVID-19." Chronic myofascial-related pain is associated with increased inflammatory mediators, serum reactive oxygen species, fibroblast growth factors, and neuroendocrine signaling (110, 111). Proinflammatory cytokines involved in COVID-19 and "long COVID-19," such as IL-1 and IL-6, may contribute to the pathogenesis of fibromyalgia (107) and low-grade fever.

5.1.6. Mood disorders

Post-traumatic stress disorder, anxiety, and depression are a manifestation of "long COVID-19" (109). Systemic fibroblast growth factors and other cytokines have an important effect on the brain (112). The intracranial ECM is suggested to have a role in neurological and psychiatric disorders (113–115). Stress (more precisely-restraint stress), arising from widespread myofibroblast-generated-tensegrity-tension in combination with psychological stress, is expected to lead to a wide range of pathologies. Interestingly, experimental models subjected to restraint stress reveal numerous abnormalities, e.g., depression and anxiety-like behavior, mechanical and cold allodynia, gut dysmotility, central nervous system abnormalities, infertility, metabolic abnormalities, and other phenomena (4, 116–125).

5.1.7. Hair loss

By the effect of the myo/fibroblast network on the connective tissue sheath and follicular papilla and on hair cycling.

"Fascial armoring" will have different manifestations depending on the anatomical location and severity of the disease and visceral organ involvement. Some symptoms will evolve over time while others will resolve on their own. ECM remodeling and stress shielding is an ongoing dynamic process. Figure 2 outlines Fascial Armoring as a neuro-biomechanical medical entity characterized by bio-tensegrity tension, imbalance, and compression with multiorgan manifestations. While this work puts forward a theoretical model for part of the mechanism of "long COVID-19," it is important to emphasize that this mechanism might be the explanation for only a subset of "long COVID-19" patients. This hypothesis and theory paper does not claim that the evidence for the suggested model is definitive or that all patients with persistent symptoms following COVID-19 have a fascial dysfunction. A connective tissue mechanism might be part of a broader mechanism or one of several parallel mechanisms occurring following COVID-19, as shown in Figure 3. Fibrotic changes and stiffening of lung tissue hinder gas exchange and result in compromised lung function leading to dyspnea and fatigue on exertion (49). Epigenetic changes are seen in "long COVID-19" and may contribute to age acceleration (126). Viral reservoirs or lingering fragments of viral particles in different tissues contribute to "long COVID-19" syndrome (72, 127).

5.2. Predicting findings in future "long COVID-19" investigations

Based on the model of myofibroblast-generated-tensegritytension and ECM alterations, predictions of results in future investigations of "long COVID-19" can be made as follows:

• Muscle damping measurements are expected to be higher in myalgic "long COVID-19" syndrome patients due to increased myofascial stiffness.

- In severe cases presenting as a fibromyalgia-like syndrome, increased intramuscular pressure is expected, depending on the site of the tensegrity pathology and imbalance. "Fascial armoring," intrinsically, has anatomical and physiological variations (4).
- Elastography studies may be used to quantify pre and post infection values of shear wave measurements and stiffness/ rigidity of fascia in different anatomical sites. Elasticity and rigidity of the fascia can be measured, and the stress-strain relationship and stress relaxation under uniaxial elongation. Load-strain and load-time curves, hysteresis, and stressrelaxation tests should find differences between severe and mild cases of "long COVID-19," and between patients and healthy controls.
- Microdialysis studies can be used to quantify signs of sub/clinical hypoxia in tissues.
- Dermatological manifestations are expected: abnormal ECM properties could predispose to easy bruising. ECM internal forces and contractility of myofibroblast populations under the skin, would lead to creeping, crawling, and itching sensations by activating various sensory receptors embedded in superficial fascia and skin. Small fiber neuropathy may be expected as well.



FIGURE 2

Outlining Fascial Armoring as a neuro-biomechanical medical entity in the form of bio-tensegrity tension and mechanical myofascioskeletal imbalance. Fascial armoring: myofibroblast-mediated-bio-tensegrity-tension and ECM abnormalities. Not all relationships are depicted with arrows in this scheme. Dashed arrows are meant for more visual clarity and have no special meaning. The occurrence of symptoms is not strictly dependent on the existence of pain. The clinical manifestation would depend on the anatomical location, structures involved, visceral organ involvement, and more. The theoretical model intrinsically has variations. ECM, extracellular matrix; POTS, postural orthostatic tachycardia syndrome. Figure created with BioRender.com, dapted from "flow chart" (2023), Available at: https://app.biorender.com/biorender.templates.



FIGURE 3

Long COVID-19 as a disease with a complex multifaceted pathophysiology that can involve multiple organ systems and tissues and an array of immune and non-immune processes. Multiple cellular mechanisms may contribute to long COVID-19. Fascial armoring is suggested to contribute to the symptomatology of long COVID-19 in cases when it manifests as a fibromyalgia-like syndrome. Although pulmonary fibrosis is mentioned, the term "long COVID-19" is generally used in the discussion to describe chronic medically unexplained symptoms after COVID-19 when no clear organic cause can be found to explain them, and with no comorbidity that can sufficiently account for them.

Besides the effects of the acute inflammatory phase of COVID-19, compression and stiffening of dermal and subcutaneous (superficial) fascial tissue could affect small neuronal fibers, perineurium, and Schwann cells. The effect of tissue stiffness and mechanical load on neuronal axons (128, 129) in this model was explained elsewhere (4)

- Ultrasound doppler can be used to measure altered blood flow in vessels. Pulse wave velocity can be used to estimate arterial stiffness. Results are expected to reflect a mild-moderate fascial compressive disorder.
- Pallor would be an overlooked manifestation in fibromyalgia-like cases. Chronic compressive forces in the periorbital fascia would lead to reduced optic disc perfusion.
- Biopsies can be taken to sample fascia from various sites, however, these affect the tensegrity structure and, if done, should be done cautiously to minimize unnecessary iatrogenic harm. Fibromyalgia patients were found to have several fascial and ECM abnormalities, including increased TGF-B gene expression in biopsy samples of peripheral fibroblasts, as well as elevated IL-1 and IL-6 levels (34, 35, 38, 41, 130, 131). Similarly, widespread disorganization of subcutaneous ECM fibers would be a "non-specific" finding on biopsies of fascia "long COVID-19." Concentration of myofascial in myofibroblasts and α -SMA should be higher. Protomyofibroblasts are an intermediate form of cells in the fibroblast-to-myofibroblast transition, and they do not contain as much α-SMA yet do generate mechanical force. Muscle biopsy from chronic cases will reveal moth-eaten appearance and ragged red fibers and changes due to low-grade chronic ischemia.

To the best of the author's knowledge, current research has yet to fully examine or study most of the above investigations in "long COVID-19."

5.3. Predicting risk factors and protective/ relieving factors

Based on the presented model, several factors are predicted to worsen/predispose or relieve "long COVID-19" and fibromyalgia-like syndromes. For example:

- Patients with hypermobility syndrome are expected to be at significantly increased risk of "long COVID-19" and "fibromyalgia." This association is predicted because collagen microarchitecture affects mechanosensitive signaling and can induce fibroblast-to-myofibroblast differentiation. Studies have shown that patients with hypermobility syndrome have widespread ECM disarray and increased myofibroblasts (4, 132). In our context, an imbalanced tensegrity structure with laxity in its frame is expected to collapse more easily.
- Sedentary and physically inactive persons (for example, during lockdowns or isolation): sedentarism is suggested to be a major factor affecting myofascial pain/fibromyalgia syndrome because immobility and disuse lead to muscle fibrosis and atrophy (4, 133).
- Inflammatory conditions contribute pro-inflammatory and pro-fibrotic input into the fascial armoring mechanism, therefore, "long COVID-19" is expected to be associated with other chronic inflammatory and autoimmune disorders.
- Individuals that tend to wear tight clothes and accessories are predicted to have a higher odds ratio/relative risk for "long COVID-19" because of the chronic mechanical input these factors provide to the myofibroblast cascade of contracture. A study suggests mechanical tension on skin induces myofibroblasts (134). According to another study, applying mechanical stimuli in the form of a splint induces myofibroblasts (135). Tight-fitting clothes are suggested to promote myofascial pain (4, 136, 137).
- Low vitamin D might predispose to disease. Vitamin D is involved in myofibroblast de-differentiation (138).

Obesity, a history of "psychosomatic" or "nonspecific" illness, and a low estrogen state (polycystic ovary syndrome, turner syndrome, post-menopausal women, men, etc.) might be predisposing factors for causing more severe or persistent disease:

- Obesity and diabetes as a risk factor may be deduced from a study (139) showing that connective tissue fibrosis induced by obesity is dependent on mechanosensitive signaling. Myofibroblast-mediated fibrosis can change the plasticity of subcutaneous tissue and increase connective tissue rigidity and stiffness (139). The effect that chronic injections have on the tensegrity system might augment the disorder.
- Under this framework, a previous medical history of chronic myofascial pain or a functional psychosomatic disorder in an individual indicates the presence of "facial armoring" in fascia and is, therefore, anticipated to be a risk factor.
- Contraceptive pills are expected to be a protective factor against "long COVID-19." It is suggested that higher incidence of fibromyalgia is not necessarily associated with abnormal estrogen levels (140, 141). Fibroblasts express hormone receptors (142, 143). Estrogen has an inhibitory effect on myofibroblast



differentiation and is also shown to be associated with lower fascial stiffness (144, 145). Hormonal contraceptives are associated with reduced risk of fibromyalgia in women (146). Estrogen receptor- β , Sirt1, and hydrogen sulfide can diminish TGF-β signaling (45). A study conducted by survey sampled 616 individuals (77.4% women) found new onset of fibromyalgia symptoms after COVID-19 infection occur more often in males (OR: 9.95, 95% CI 6.02 to 16.43, *p* < 0.0001) and in obesity (OR: 41.20, 95% CI 18.00 to 98.88, *p* < 0.0001) (107), although one should take into account that online surveys have intrinsic limitations and biases such as self-selection bias (107). The sex distribution of "long COVID-19" would depend on estrogen levels as well as multiple other factors including lifestyle factors and other (genetic, environmental, behavioral, etc.). Interestingly, with obesity being such a strong associated predictor of fibromyalgia after COVID-19 (OR>40), the intensity of treatment setting had odds ratio of only ~1.7, pre-existing anxiety had an odds ratio of ~1.3 (95% CI 0.858 to 2.013), and pre-existing depression had odds ratio of 1.255, p-value=0.507. Meanwhile, neurobiopsychosocial theories for fibromyalgia put a major emphasis on stress and psychology.

Figure 4 illustrates various inputs into the positive feedback loop of fascial armoring, where factors that stimulate or inhibit

myofibroblasts provide their input. The theory of "fascial armoring" links between unhealthy lifestyle (e.g., diet, sedentarism, smoking etc.) and pain and suffering. As with most conditions, it is multifactorial. Lifestyle interventions and exercise (movement), therefore, may have a major role in the treatment of "long COVID-19" and in psychosomatic syndromes. A study showed cyclical mechanical stress reduces myofibroblast differentiation of primary lung fibroblasts (147). Movement might help release fascial tensions by various mechanisms (4) and immobility leads to fibrosis and increase in myofibroblasts (4, 133). Needling therapy (when done correctly) is expected to modulate the disease course because of its suggested effects on the bio-tensegrity system (7).

6. Conclusion

"Long COVID-19" exhibits fibromyalgia-like manifestations and symptomatology including chronic fatigue, cognitive impairment, low mood, functional impairment, and last but not least-myofascial pain (i.e., "fibromyalgia-ness"). Globally, the long COVID-19 wave has risen and is leading to extensive morbidity, decreased quality of life, and suffering in society. SARS-CoV-2 infection can lead to pro-inflammatory and pro-fibrotic responses in multiple tissues, including myofascial tissue, and leads to fibrosis and upregulation of

myofibroblasts. Multiple parallel mechanisms might be at work in long COVID-19 syndrome, which is hypothesized in this manuscript to involve a chronic connective tissue pathology and changes in the myofascioskeletal bio-tensegrity system as well as immune dysregulation, viral persistence, immune-mediated myopathy, muscle atrophy, and other abnormalities. Further research is required to enhance our understanding of the underlying mechanisms of- and the extent of soft tissue damage in- "long COVID-19 syndrome." The coronavirus pandemic offers us an opportunity to better understand peripheral mechanisms of "functional psycho/somatic syndromes" such as "fibromyalgia," "ME/CFS," "post Lyme disease," and related conditions. The boundary of a medical entity is determined by the boundary of the biological process that underlies it. This theoretical model offers one mechanical aspect as part of a common immuno-rheumo-psychoneurological mechanism. Biology does not separate itself into different medical specialties, and the body and the mind are one flesh, one Being. The principle of Occam's razor (i.e., "medical monotheism") favors minimizing the multiplication of entities. According to fascial armoring, indeed, there is only one entity...

Data availability statement

The original contributions presented in the study are included in the article/supplementary material, further inquiries can be directed to the corresponding author.

Author contributions

SP solely contributed to the study conception and design, material preparation, data collection, data curation, conceptualization,

References

1. Crook H, Raza S, Nowell J, Young M, Edison P. Long covid-mechanisms, risk factors, and management. *BMJ*. (2021) 374:n1648. doi: 10.1136/bmj.n1648

2. Colarusso C, Maglio A, Terlizzi M, Vitale C, Molino A, Pinto A, et al. Post-COVID-19 patients who develop lung fibrotic-like changes have lower circulating levels of IFN- β but higher levels of IL-1 α and TGF- β . *Biomedicine*. (2021) 9:1931. doi: 10.3390/biomedicines9121931

3. Proal AD, VanElzakker MB. Long COVID or post-acute sequelae of COVID-19 (PASC): an overview of biological factors that may contribute to persistent symptoms. *Front Microbiol.* (2021) 12:698169. doi: 10.3389/fmicb.2021.698169

4. Plaut S. Scoping review and interpretation of myofascial pain/fibromyalgia syndrome: an attempt to assemble a medical puzzle. *PLoS One.* (2022) 17:e0263087. doi: 10.1371/journal.pone.0263087

5. Firestein GS, Kelley WN. *Kelley's textbook of rheumatology. 9th* Edn. Philadelphia, PA: Elsevier/Saunders (2013). pp. 733–750.

6. Tadeo I, Berbegall AP, Escudero LM, Álvaro T, Noguera R. Biotensegrity of the extracellular matrix: physiology, dynamic mechanical balance, and implications in oncology and mechanotherapy. *Front Oncol.* (2014) 4:39. doi: 10.3389/ fonc.2014.00039

7. Plaut S. Suggesting a mechanism for acupuncture as a global percutaneous needle fasciotomy that respects tensegrity principles for treating fibromyalgia. *Front Med.* (2023) 9:952159. doi: 10.3389/fmed.2022.952159

 Caldeira P, Fonseca ST, Paulo A, Infante J, Araújo D. Linking tensegrity to sports team collective behaviors: towards the group-tensegrity hypothesis. *Sports Med Open*. (2020) 6:24. doi: 10.1186/s40798-020-00253-y

9. Wilke J, Krause F, Vogt L, Banzer W. What is evidence-based about myofascial chains: a systematic review. *Arch Phys Med Rehabil.* (2016) 97:454–61. doi: 10.1016/j. apmr.2015.07.023

integration of information, formal analysis investigation, methodology, visualization, resources, writing—original draft, and writing—review and editing.

Acknowledgments

The author would like to thank Beezrat Hashem, Hakadosh Baruch Hoo, Elohei Col Beriyot, Adon Col Toladot, Ribon Col Hamaasim, Ha-Mefaaneach Neelamim, Messiach Ilmim, Pokeach Ivrim, Zokef Kefufim, Mechaye Metim, and Rofe Cholim, for assisting with supervision and guidance; Pnina Ohanna Plaut for assisting with editing and aiding the author's academic apprenticeship; and Yona Alef Tet, Ivri Anochi, Ve-et Adonai, Elohei Hashamayim, Ani Yare, Asher Asa, Et Hayam, Ve-et Hayabasha. Mishlei Perek Alef, for sharing wisdom.

Conflict of interest

The author declares that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

10. Micheletti A, Podio-Guidugli P. Seventy years of tensegrities (and counting). Arch Appl Mech. (2022) 92:2525–48. doi: 10.1007/s00419-022-02192-4

11. Scarr G. Biotensegrity: what is the big deal? J Bodyw Mov Ther. (2020) 24:134–7. doi: 10.1016/j.jbmt.2019.09.006

12. Zheng N, Yuan XY, Chi YY, Liu P, Wang B, Sui JY, et al. The universal existence of myodural bridge in mammals: an indication of a necessary function. *Sci Rep.* (2017) 7:8248. doi: 10.1038/s41598-017-06863-z

13. Tomasek JJ, Gabbiani G, Hinz B, Chaponnier C, Brown RA. Myofibroblasts and mechano-regulation of connective tissue remodelling. *Nat Rev Mol Cell Biol.* (2002) 3:349–63. doi: 10.1038/nrm809

14. Stempien-Otero A, Kim DH, Davis J. Molecular networks underlying myofibroblast fate and fibrosis. *J Mol Cell Cardiol.* (2016) 97:153–61. doi: 10.1016/j. yjmcc.2016.05.002

15. Majno G, Gabbiani G, Hirschel BJ, Ryan GB, Statkov PR. Contraction of granulation tissue in vitro: similarity to smooth muscle. *Science*. (1971) 173:548–50. doi: 10.1126/science.173.3996.548

16. Xie T, Wang Y, Deng N, Huang G, Taghavifar F, Geng Y, et al. Single-cell deconvolution of fibroblast heterogeneity in mouse pulmonary fibrosis. *Cell Rep.* (2018) 22:3625–40. doi: 10.1016/j.celrep.2018.03.010

17. Stecco C, Fede C, Macchi V, Porzionato A, Petrelli L, Biz C, et al. The fasciacytes: a new cell devoted to fascial gliding regulation. *Clin Anat.* (2018) 31:667–76. doi: 10.1002/ca.23072

18. Zhao W, Wang X, Sun KH, Zhou L. α -Smooth muscle actin is not a marker of fibrogenic cell activity in skeletal muscle fibrosis. *PLoS One.* (2018) 13:e0191031. doi: 10.1371/journal.pone.0191031

19. Schleip R, Klingler W. Active contractile properties of fascia. *Clin Anat.* (2019) 32:891–5. doi: 10.1002/ca.23391

20. Fede C, Pirri C, Fan C, Petrelli L, Guidolin D, De Caro R, et al. A closer look at the cellular and molecular components of the deep/muscular fasciae. *Int J Mol Sci.* (2021) 22:1411. doi: 10.3390/ijms22031411

21. Hansson E, Skiöldebrand E. Coupled cell networks are target cells of inflammation, which can spread between different body organs and develop into systemic chronic inflammation. *J Inflamm*. (2015) 12:44. doi: 10.1186/s12950-015-0091-2

22. Langevin HM, Cornbrooks CJ, Taatjes DJ. Fibroblasts form a body-wide cellular network. *Histochem Cell Biol.* (2004) 122:7–15. doi: 10.1007/s00418-004-0667-z

23. Fede C, Porzionato A, Petrelli L, Fan C, Pirri C, Biz C, et al. Fascia and soft tissues innervation in the human hip and their possible role in post-surgical pain. *J Orthop Res.* (2020) 38:1646–54. doi: 10.1002/jor.24665

24. Sinhorim L, Amorim MDS, Ortiz ME, Bittencourt EB, Bianco G, da Silva FC, et al. Potential nociceptive role of the thoracolumbar fascia: a scope review involving in vivo and ex vivo studies. *J Clin Med.* (2021) 10:4342. doi: 10.3390/jcm10194342

25. Katz RS, Leavitt F, Small AK, Small AK, Small BJ. Intramuscular pressure is almost three times higher in fibromyalgia patients: a possible mechanism for understanding the muscle pain and tenderness. *J Rheumatol.* (2020) 48:955. doi: 10.3899/jrheum.191068

26. Costanzo LS. Physiology (6th Edn.). Philadelphia, PA: Elsevier (2018). pp. 128-129.

27. Meehan William P., O'Brien Michael J.. (2021) "chronic exertional compartment syndrome" (section performance and interpretation of testing). Available at: https://www.wolterskluwer.com/en/solutions/uptodate (Accessed August 2021).

28. Morf S, Amann-Vesti B, Forster A, Franzeck UK, Koppensteiner R, Uebelhart D, et al. Microcirculation abnormalities in patients with fibromyalgia—measured by capillary microscopy and laser fluxmetry. *Arthritis Res Ther.* (2005) 7:R209–16. doi: 10.1186/ar1459

29. Grassi W, Core P, Carlino G, Salaffi F, Cervini C. Capillary permeability in fibromyalgia. J Rheumatol. (1994) 21:1328-31.

30. Triantafyllias K, Stortz M, de Blasi M, Leistner C, Weinmann-Menke J, Schwarting A. Increased aortic stiffness in patients with fibromyalgia: results of a prospective study on carotid-femoral pulse wave velocity. *Clin Exp Rheumatol.* (2019) 116:114–5.

31. Kalyan-Raman UP, Kalyan-Raman K, Yunus MB, Masi AT. Muscle pathology in primary fibromyalgia syndrome: a light microscopic, histochemical and ultrastructural study. *J Rheumatol.* (1984) 11:808–13.

32. Wachter KC, Kaeser HE, Gühring H, Ettlin TM, Mennet P, Müller W. Muscle damping measured with a modified pendulum test in patients with fibromyalgia, lumbago, and cervical syndrome. *Spine*. (1996) 21:2137–42. doi: 10.1097/00007632-199609150-00016

33. Sprott H, Salemi S, Gay RE, Bradley LA, Alarcón GS, Oh SJ, et al. Increased DNA fragmentation and ultrastructural changes in fibromyalgic muscle fibres. *Ann Rheum Dis.* (2004) 63:245–51. doi: 10.1136/ard.2002.004762

34. Evdokimov D, Kreß L, Dinkel P, Frank J, Sommer C, Üçeyler N. Pain-associated mediators and axon pathfinders in fibromyalgia skin cells. *J Rheumatol.* (2020) 47:140–8. doi: 10.3899/jrheum.190248

35. Salemi S, Rethage J, Wollina U, Michel BA, Gay RE, Gay S, et al. Detection of interleukin 1beta (IL-1beta), IL-6, and tumor necrosis factor-alpha in skin of patients with fibromyalgia. *J Rheumatol.* (2003) 30:146–50.

36. Henderson NC, Rieder F, Wynn TA. Fibrosis: from mechanisms to medicines. *Nature*. (2020) 587:555–66. doi: 10.1038/s41586-020-2938-9

37. O'Mahony LF, Srivastava A, Mehta P, Ciurtin C. Is fibromyalgia associated with a unique cytokine profile? A systematic review and meta-analysis. *Rheumatology*. (2021) 60:2602–14. doi: 10.1093/rheumatology/keab146

38. Ramírez-Tejero JA, Martínez-Lara E, Peinado MÁ, Del MML, Siles E. Hydroxytyrosol as a promising ally in the treatment of fibromyalgia. *Nutrients*. (2020) 12:2386. doi: 10.3390/nu12082386

39. Gerdle B, Söderberg K, Puigvert LS, Rosendal L, Larsson B. Increased interstitial concentrations of pyruvate and lactate in the trapezius muscle of patients with fibromyalgia: a microdialysis study. *J Rehabil Med.* (2010) 42:679–87. doi: 10.2340/16501977-0581

40. McIver KL, Evans C, Kraus RM, Ispas L, Sciotti VM, Hickner RC. NO-mediated alterations in skeletal muscle nutritive blood flow and lactate metabolism in fibromyalgia. *Pain.* (2006) 120:161–9. doi: 10.1016/j.pain.2005.10.032

41. Sprott H, Müller A, Heine H. Collagen crosslinks in fibromyalgia. Arthritis Rheum. (1997) 40:1450–4. doi: 10.1002/art.1780400813

42. Gronemann ST, Ribel-Madsen S, Bartels EM, Danneskiold-Samsoe B, Bliddal H. Collagen and muscle pathology in fibromyalgia patients. *Rheumatology*. (2004) 43:27–31. doi: 10.1093/rheumatology/keg452

43. Fede C, Petrelli L, Guidolin D, Porzionato A, Pirri C, Fan C, et al. Evidence of a new hidden neural network into deep fasciae. *Sci Rep.* (2021) 11:12623. doi: 10.1038/s41598-021-92194-z

44. Levy Nogueira M, Hamraz M, Abolhassani M, Bigan E, Lafitte O, Steyaert JM, et al. Mechanical stress increases brain amyloid β , tau, and α -synuclein concentrations in wild-type mice. *Alzheimers Dement.* (2018) 14:444–53. doi: 10.1016/j.jalz.2017.11.003

45. Dinicolantonio JJ, McCarty MF, Barroso-Aranda J, Assanga S, Lujan LML, O'Keefe JH. A nutraceutical strategy for downregulating TGF β signalling: prospects for prevention of fibrotic disorders, including post-COVID-19 pulmonary fibrosis. *Open Heart.* (2021) 8:e001663. doi: 10.1136/openhrt-2021-001663

46. Meng XM, Nikolic-Paterson DJ, Lan HY. TGF-β: the master regulator of fibrosis. Nat Rev Nephrol. (2016) 12:325–38. doi: 10.1038/nrneph.2016.48

47. Giacomelli C, Piccarducci R, Marchetti L, Romei C, Martini C. Pulmonary fibrosis from molecular mechanisms to therapeutic interventions: lessons from post-COVID-19 patients. *Biochem Pharmacol.* (2021) 193:114812. doi: 10.1016/j.bcp.2021.114812

48. Hartmann C, Miggiolaro Afr Dos S, Motta J Da S, Baena Carstens L, Busatta Vaz De Paula C, Fagundes Grobe S, et al. The pathogenesis of COVID-19 myocardial injury: an Immunohistochemical study of postmortem biopsies. *Front Immunol.* (2021) 12:748417. doi: 10.3389/fimmu.2021.748417

49. John AE, Joseph C, Jenkins G, Tatler AL. COVID-19 and pulmonary fibrosis: a potential role for lung epithelial cells and fibroblasts. *Immunol Rev.* (2021) 302:228–40. doi: 10.1111/imr.12977

50. Martin K, Pritchett J, Llewellyn J, Mullan AF, Athwal VS, Dobie R, et al. PAK proteins and YAP-1 signalling downstream of integrin beta-1 in myofibroblasts promote liver fibrosis. *Nat Commun.* (2016) 7:12502. doi: 10.1038/ncomms12502

51. Ramazani Y, Knops N, Elmonem MA, Nguyen TQ, Arcolino FO, van den Heuvel L, et al. Connective tissue growth factor (CTGF) from basics to clinics. *Matrix Biol.* (2018) 68-69:44-66. doi: 10.1016/j.matbio.2018.03.007

52. Sugiura H, Ichikawa T, Koarai A, Yanagisawa S, Minakata Y, Matsunaga K, et al. Activation of toll-like receptor 3 augments myofibroblast differentiation. *Am J Respir Cell Mol Biol.* (2009) 40:654–62. doi: 10.1165/rcmb.2008-0371OC

53. Koziczak-Holbro M, Joyce C, Glück A, Kinzel B, Müller M, Tschopp C, et al. IRAK-4 kinase activity is required for interleukin-1 (IL-1) receptor- and toll-like receptor 7-mediated signaling and gene expression. *J Biol Chem.* (2007) 282:13552–60. doi: 10.1074/jbc.M700548200

54. Pandolfi L, Bozzini S, Frangipane V, Percivalle E, de Luigi A, Violatto MB, et al. Neutrophil extracellular traps induce the epithelial-mesenchymal transition: implications in post-COVID-19 fibrosis. *Front Immunol.* (2021) 12:663303. doi: 10.3389/ fimmu.2021.663303

55. Wang S, Meng XM, Ng YY, Ma FY, Zhou S, Zhang Y, et al. TGF-β/Smad3 signalling regulates the transition of bone marrow-derived macrophages into myofibroblasts during tissue fibrosis. *Oncotarget*. (2016) 7:8809–22. doi: 10.18632/oncotarget.6604

56. Gracia-Hernandez M, Sotomayor EM, Villagra A. Targeting macrophages as a therapeutic option in coronavirus disease 2019. *Front Pharmacol.* (2020) 11:577571. doi: 10.3389/fphar.2020.577571

57. Datta SD, Talwar A, Lee JT. A proposed framework and timeline of the Spectrum of disease due to SARS-CoV-2 infection: illness beyond acute infection and public health implications. *JAMA*. (2020) 324:2251–2. doi: 10.1001/jama.2020.22717

58. Townsend L, Dowds J, O'Brien K, Sheill G, Dyer AH, O'Kelly B, et al. Persistent poor health after COVID-19 is not associated with respiratory complications or initial disease severity. *Ann Thorac Soc.* (2021) 18:997–1003. doi: 10.1513/ AnnalsATS.202009-1175OC

59. Zhou M, Xu J, Liao T, Yin Z, Yang F, Wang K, et al. Comparison of residual pulmonary abnormalities 3 months after discharge in patients who recovered from COVID-19 of different severity. *Front Med.* (2021) 8:682087. doi: 10.3389/fmed.2021.682087

60. Zhou F, Tao M, Shang L, Liu Y, Pan G, Jin Y, et al. Assessment of sequelae of COVID-19 nearly 1 year after diagnosis. *Front Med.* (2021) 8:717194. doi: 10.3389/fmed.2021.717194

61. Chun HJ, Coutavas E, Pine AB, Lee AI, Yu VL, Shallow MK, et al. Immunofibrotic drivers of impaired lung function in postacute sequelae of SARS-CoV-2 infection. *JCI Insight*. (2021) 6:e148476. doi: 10.1172/jci.insight.148476

62. Mikkelsen Mark E., Abramoff Benjamin. (2022). "COVID-19: Evaluation and management of adults following acute viral illness" section: Persistent symptoms. Available at: https://www.wolterskluwer.com/en/solutions/uptodate (Accessed February 2022).

63. Lauwers M, Au M, Yuan S, Wen C. COVID-19 in joint ageing and osteoarthritis: current status and perspectives. *Int J Mol Sci.* (2022) 23:720. doi: 10.3390/ijms23020720

64. Willi S, Lüthold R, Hunt A, Hänggi NV, Sejdiu D, Scaff C, et al. COVID-19 sequelae in adults aged less than 50 years: a systematic review. *Travel Med Infect Dis.* (2021) 40:101995. doi: 10.1016/j.tmaid.2021.101995

65. Davis HE, Assaf GS, McCorkell L, Wei H, Low RJ, Reem Y, et al. Characterizing long COVID in an international cohort: 7 months of symptoms and their impact. *EClinicalMedicine*. (2021) 38:101019. doi: 10.1016/j.eclinm.2021.101019

66. Melms JC, Biermann J, Huang H, Wang Y, Nair A, Tagore S, et al. A molecular single-cell lung atlas of lethal COVID-19. *Nature*. (2021) 595:114–9. doi: 10.1038/s41586-021-03569-1

67. Yao XH, Luo T, Shi Y, He ZC, Tang R, Zhang PP, et al. A cohort autopsy study defines COVID-19 systemic pathogenesis. *Cell Res.* (2021) 31:836–46. doi: 10.1038/s41422-021-00523-8

68. Omer D, Pleniceanu O, Gnatek Y, Namestnikov M, Cohen-Zontag O, Goldberg S, et al. Human kidney spheroids and monolayers provide insights into SARS-CoV-2 renal interactions. *J Am Soc Nephrol.* (2021) 32:2242–54. doi: 10.1681/ASN.2020111546

69. Deshmukh V, Motwani R, Kumar A, Kumari C, Raza K. Histopathological observations in COVID-19: a systematic review. *J Clin Pathol.* (2021) 74:76–83. doi: 10.1136/jclinpath-2020-206995

70. Beydon M, Chevalier K, Al Tabaa O, Hamroun S, Delettre AS, Thomas M, et al. Myositis as a manifestation of SARS-CoV-2. *Ann Rheum Dis.* (2021) 80:e42. doi: 10.1136/annrheumdis-2020-217573

71. Smelcerovic A, Kocic G, Gajic M, Tomovic K, Djordjevic V, Stankovic-Djordjevic D, et al. DPP-4 inhibitors in the prevention/treatment of pulmonary fibrosis, heart and kidney injury caused by COVID-19-a therapeutic approach of choice in type 2 diabetic patients? *Front Pharmacol.* (2020) 11:1185. doi: 10.3389/fphar.2020.01185

72. Cenko E, Badimon L, Bugiardini R, Claeys MJ, De Luca G, de Wit C, et al. Cardiovascular disease and COVID-19: a consensus paper from the ESC working group on Coronary Pathophysiology & Microcirculation, ESC working group on thrombosis and the Association for Acute CardioVascular care (ACVC), in collaboration with the European heart rhythm association (EHRA). *Cardiovasc Res.* (2021) 117:2705–29. doi: 10.1093/cvr/cvab298

73. Unudurthi SD, Luthra P, Bose RJC, McCarthy JR, Kontaridis MI. Cardiac inflammation in COVID-19: lessons from heart failure. *Life Sci.* (2020) 260:118482. doi: 10.1016/j.lfs.2020.118482

74. Zickler M, Stanelle-Bertram S, Ehret S, Heinrich F, Lange P, Schaumburg B, et al. Replication of SARS-CoV-2 in adipose tissue determines organ and systemic lipid metabolism in hamsters and humans. *Cell Metab.* (2022) 34:1–2. doi: 10.1016/j. cmet.2021.12.002

75. Kolesova O, Vanaga I, Laivacuma S, Derovs A, Kolesovs A, Radzina M, et al. Intriguing findings of liver fibrosis following COVID-19. *BMC Gastroenterol.* (2021) 21:370. doi: 10.1186/s12876-021-01939-7

76. Esposito G, Pesce M, Seguella L, Sanseverino W, Lu J, Sarnelli G. Can the enteric nervous system be an alternative entrance door in SARS-CoV2 neuroinvasion? *Brain Behav Immun.* (2020) 87:93–4. doi: 10.1016/j.bbi.2020.04.060

77. Hoffmann M, Kleine-Weber H, Schroeder S, Krüger N, Herrler T, Erichsen S, et al. SARS-CoV-2 cell entry depends on ACE2 and TMPRSS2 and is blocked by a clinically proven protease inhibitor. *Cells.* (2020) 181:271–280.e8. doi: 10.1016/j.cell.2020.02.052

78. Queiroz-Junior CM, Santos ACPM, Galvão I, Souto GR, Mesquita RA, Sá MA, et al. The angiotensin converting enzyme 2/angiotensin-(1-7)/mas receptor axis as a key player in alveolar bone remodeling. *Bone*. (2019) 128:115041. doi: 10.1016/j. bone.2019.115041

79. Sapra L, Saini C, Garg B, Gupta R, Verma B, Mishra PK, et al. Long-term implications of COVID-19 on bone health: pathophysiology and therapeutics. *Inflamm Res.* (2022) 71:1025–40. doi: 10.1007/s00011-022-01616-9

80. Mokuda S, Tokunaga T, Masumoto J, Sugiyama E. Angiotensin-converting enzyme 2, a SARS-CoV-2 receptor, is upregulated by interleukin 6 through STAT3 signaling in synovial tissues. *J Rheumatol.* (2020) 47:1593–5. doi: 10.3899/jrheum.200547

81. Disser NP, De Micheli AJ, Schonk MM, Konnaris MA, Piacentini AN, Edon DL, et al. Musculoskeletal Consequences of COVID-19. *J Bone Joint Surg Am.* (2020) 102:1197–204. doi: 10.2106/JBJS.20.00847

82. Hamming I, Timens W, Bulthuis ML, Lely AT, Navis G, van Goor H. Tissue distribution of ACE2 protein, the functional receptor for SARS coronavirus. A first step in understanding SARS pathogenesis. *J Pathol.* (2004) 203:631–7. doi: 10.1002/path.1570

83. Grobe JL, Der Sarkissian S, Stewart JM, Meszaros JG, Raizada MK, Katovich MJ. ACE2 overexpression inhibits hypoxia-induced collagen production by cardiac fibroblasts. *Clin Sci.* (2007) 113:357–64. doi: 10.1042/CS20070160

84. Dong M, Zhang J, Ma X, Tan J, Chen L, Liu S, et al. ACE2, TMPRSS2 distribution and extrapulmonary organ injury in patients with COVID-19. *Biomed Pharmacother*. (2020) 131:110678. doi: 10.1016/j.biopha.2020.110678. Epub 2020 Aug 24

85. Saba L, Gerosa C, Fanni D, Marongiu F, La Nasa G, Caocci G, et al. Molecular pathways triggered by COVID-19 in different organs: ACE2 receptor-expressing cells under attack? A review. *Eur Rev Med Pharmacol Sci.* (2020) 24:12609–22. doi: 10.26355/eurrev_202012_24058

86. Reiterer M, Rajan M, Gomez-Banoy N, Lau JD, Gómez-Escobar LG, Gilani A, et al. Hyperglycemia in acute COVID-19 is characterized by insulin resistance and adipose tissue infectivity by SARS-CoV-2. *Cell Metab.* (2021) 33:2174–2188.e5. doi: 10.1016/j. cmet.2021.09.009

87. Shin J, Toyoda S, Nishitani S, Fukuhara A, Kita S, Otsuki M, et al. Possible involvement of adipose tissue in patients with older age, obesity, and diabetes with SARS-CoV-2 infection (COVID-19) via GRP78 (BIP/HSPA5): significance of hyperinsulinemia management in COVID-19. *Diabetes*. (2021) 70:2745–55. doi: 10.2337/db20-1094

88. Kruglikov IL, Scherer PE. The role of adipocytes and adipocyte-like cells in the severity of COVID-19 infections. *Obesity*. (2020) 28:1187–90. doi: 10.1002/oby.22856

89. Anand H, Ende V, Singh G, Qureshi I, Duong TQ, Mehler MF. Nervous systemsystemic crosstalk in SARS-CoV-2/COVID-19: a unique Dyshomeostasis syndrome. *Front Neurosci.* (2021) 15:727060. doi: 10.3389/fnins.2021.727060 90. Xu J, Xu X, Jiang L, Dua K, Hansbro PM, Liu G. SARS-CoV-2 induces transcriptional signatures in human lung epithelial cells that promote lung fibrosis. *Respir Res.* (2020) 21:182. doi: 10.1186/s12931-020-01445-6

91. Soares MN, Eggelbusch M, Naddaf E, Gerrits KHL, van der Schaaf M, van den Borst B, et al. Skeletal muscle alterations in patients with acute Covid-19 and post-acute sequelae of Covid-19. *J Cachexia Sarcopenia Muscle*. (2022) 13:11–22. doi: 10.1002/jcsm.12896

92. Shi Z, de Vries HJ, Vlaar APJ, van der Hoeven J, Boon RA, Heunks LMA, et al. Diaphragm pathology in critically ill patients with COVID-19 and postmortem findings from 3 medical centers. *JAMA Intern Med.* (2021) 181:122–4. doi: 10.1001/jamainternmed.2020.6278

93. Leung TW, Wong KS, Hui AC, To KFLai ST, Ng WF, et al. Myopathic changes associated with severe acute respiratory syndrome: a postmortem case series. *Arch Neurol.* (2005) 62:1113–7. doi: 10.1001/archneur.62.7.1113

94. Veronesi F, Contartese D, Martini L, Visani A, Fini M. Speculation on the pathophysiology of musculoskeletal injury with COVID-19 infection. *Front Med.* (2022) 9:930789. doi: 10.3389/fmed.2022.930789

95. Aschman T, Schneider J, Greuel S, Meinhardt J, Streit S, Goebel HH, et al. Association between SARS-CoV-2 infection and immune-mediated myopathy in patients who have died. *JAMA Neurol.* (2021) 78:948–60. doi: 10.1001/jamaneurol.2021.2004

96. Suh J, Mukerji SS, Collens SI, Padera RF Jr, Pinkus GS, Amato AA, et al. Skeletal muscle and peripheral nerve histopathology in COVID-19. *Neurology*. (2021) 97:e849–58. doi: 10.1212/WNL.000000000012344

97. Patel J, Javed S. Myofascial pain syndrome and SARS-CoV-2: a case series. *Pain Manag.* (2022) 12:255–60. doi: 10.2217/pmt-2021-0044

98. Zha M, Chaffee K, Alsarraj J. Trigger point injections and dry needling can be effective in treating long COVID syndrome-related myalgia: a case report. *J Med Case Rep.* (2022) 16:31. doi: 10.1186/s13256-021-03239-w

99. Grehl C, Schultheiß C, Hoffmann K, Binder M, Altmann T, Grosse I, et al. Detection of SARS-CoV-2 derived Small RNAs and changes in circulating Small RNAs associated with COVID-19. *Viruses.* (2021) 13:1593. doi: 10.3390/v13081593

100. Komaroff AL, Lipkin WI. Insights from myalgic encephalomyelitis/chronic fatigue syndrome may help unravel the pathogenesis of postacute COVID-19 syndrome. *Trends Mol Med.* (2021) 27:895–906. doi: 10.1016/j.molmed.2021.06.002

101. Dillon GA, Wolf ST, Alexander LM. Nitric oxide-mediated cutaneous microvascular function is not altered in young adults following mild-to-moderate SARS CoV-2 infection. *Am J Physiol Heart Circ Physiol.* (2022) 322:H319–27. doi: 10.1152/ajpheart.00602.2021

102. Nandadeva D, Young BE, Stephens BY, Grotle AK, Skow RJ, Middleton AJ, et al. Blunted peripheral but not cerebral vasodilator function in young otherwise healthy adults with persistent symptoms following COVID-19. *Am J Physiol Heart Circ Physiol.* (2021) 321:H479–84. doi: 10.1152/ajpheart.00368.2021

103. Ratchford SM, Stickford JL, Province VM, Stute N, Augenreich MA, Koontz LK, et al. Vascular alterations among young adults with SARS-CoV-2. *Am J Physiol Heart Circ Physiol*. (2021) 320:H404–10. doi: 10.1152/ajpheart.00897.2020

104. Jud P, Gressenberger P, Muster V, Avian A, Meinitzer A, Strohmaier H, et al. Evaluation of endothelial dysfunction and inflammatory vasculopathy after SARS-CoV-2 infection-a cross-sectional study. *Front Cardiovasc Med.* (2021) 8:750887. doi: 10.3389/fcvm.2021.750887

105. Eapen MS, Lu W, Gaikwad AV, Bhattarai P, Chia C, Hardikar A, et al. Endothelial to mesenchymal transition: a precursor to post-COVID-19 interstitial pulmonary fibrosis and vascular obliteration? *Eur Respir J.* (2020) 56:2003167. doi: 10.1183/13993003.03167-2020

106. Yasui M, Menjyo Y, Tokizane K, Shiozawa A, Tsuda M, Inoue K, et al. Hyperactivation of proprioceptors induces microglia-mediated long-lasting pain in a rat model of chronic fatigue syndrome. *J Neuroinflammation*. (2019) 16:67. doi: 10.1186/ s12974-019-1456-x

107. Ursini F, Ciaffi J, Mancarella L, Lisi L, Brusi V, Cavallari C, et al. Fibromyalgia: a new facet of the post-COVID-19 syndrome spectrum? Results from a web-based survey. *RMD Open*. (2021) 7:e001735. doi: 10.1136/rmdopen-2021-001735

108. Bileviciute-Ljungar I, Norrefalk JR, Borg K. Pain burden in post-COVID-19 syndrome following mild COVID-19 infection. *J Clin Med.* (2022) 11:771. doi: 10.3390/jcm11030771

109. Andrade BS, Siqueira S, de Assis Soares WR, de Souza RF, Santos NO, dos Santos FA, et al. Long-COVID and post-COVID health complications: an up-to-date review on clinical conditions and their possible molecular mechanisms. *Viruses.* (2021) 13:700. doi: 10.3390/v13040700

110. Grosman-Rimon L, Parkinson W, Upadhye S, Clarke H, Katz J, Flannery J, et al. Circulating biomarkers in acute myofascial pain: a case-control study. *Medicine*. (2016) 95:e4650. doi: 10.1097/MD.000000000004650

111. Basi DL, Velly AM, Schiffman EL, Lenton PA, Besspiata DA, Rankin AM, et al. Human temporomandibular joint and myofascial pain biochemical profiles: a casecontrol study. *J Oral Rehabil.* (2012) 39:326–37. doi: 10.1111/j.1365-2842.2011.02271.x

112. Katsouri L, Ashraf A, Birch AM, Lee KK, Mirzaei N, Sastre M. Systemic administration of fibroblast growth factor-2 (FGF2) reduces BACE1 expression and

amyloid pathology in APP23 mice. Neurobiol Aging. (2015) 36:821-31. doi: 10.1016/j. neurobiolaging.2014.10.004

113. Koskinen MK, van Mourik Y, Smit AB, Riga D, Spijker S. From stress to depression: development of extracellular matrix-dependent cognitive impairment following social stress. *Sci Rep.* (2020) 10:17308. doi: 10.1038/s41598-020-73173-2

114. Dityatev A, Seidenbecher C, Morawski M. Brain extracellular matrix: an upcoming target in neurological and psychiatric disorders. *Eur J Neurosci.* (2021) 53:3807–10. doi: 10.1111/ejn.15336

115. Sun Y, Xu S, Jiang M, Liu X, Yang L, Bai Z, et al. Role of the extracellular matrix in Alzheimer's disease. *Front Aging Neurosci.* (2021) 13:707466. doi: 10.3389/fnagi.2021.707466

116. Deng Y, Zhou M, Wang J, Yao J, Yu J, Liu W, et al. Involvement of the microbiotagut-brain axis in chronic restraint stress: disturbances of the kynurenine metabolic pathway in both the gut and brain. *Gut Microbes*. (2021) 13:1–16. doi: 10.1080/19490976.2020.1869501

117. Chiba S, Numakawa T, Ninomiya M, Richards MC, Wakabayashi C, Kunugi H. Chronic restraint stress causes anxiety- and depression-like behaviors, downregulates glucocorticoid receptor expression, and attenuates glutamate release induced by brainderived neurotrophic factor in the prefrontal cortex. *Prog Neuro-Psychopharmacol Biol Psychiatry*. (2012) 39:112–9. doi: 10.1016/j.pnpbp.2012.05.018

118. Farhin S, Wong A, Delungahawatta T, Amin JY, Bienenstock J, Buck R, et al. Restraint stress induced gut dysmotility is diminished by a milk oligosaccharide (2'-fucosyllactose) in vitro. *PLoS One.* (2019) 14:e0215151. doi: 10.1371/journal. pone.0215151

119. Wu XF, Yuan HJ, Li H, Gong S, Lin J, Miao YL, et al. Restraint stress on female mice diminishes the developmental potential of oocytes: roles of chromatin configuration and histone modification in germinal vesicle stage oocytes. *Biol Reprod.* (2015) 92:13. doi: 10.1095/biolreprod.114.124396. Epub 2014 Nov 19

120. Sugama S, Sekiyama K, Kodama T, Takamatsu Y, Takenouchi T, Hashimoto M, et al. Chronic restraint stress triggers dopaminergic and noradrenergic neurodegeneration: possible role of chronic stress in the onset of Parkinson's disease. *Brain Behav Immun.* (2016) 51:39–46. doi: 10.1016/j.bbi.2015.08.015

121. Bardin L, Malfetes N, Newman-Tancredi A, Depoortère R. Chronic restraint stress induces mechanical and cold allodynia, and enhances inflammatory pain in rat: relevance to human stress-associated painful pathologies. *Behav Brain Res.* (2009) 205:360–6. doi: 10.1016/j.bbr.2009.07.005

122. Liu N, Wang LH, Guo LL, Wang GQ, Zhou XP, Jiang Y, et al. Chronic restraint stress inhibits hair growth via substance P mediated by reactive oxygen species in mice. *PLoS One.* (2013) 8:e61574. doi: 10.1371/journal.pone.0061574

123. Spyrka J, Hess G. Repeated neck restraint stress Bidirectionally modulates excitatory transmission in the dentate gyrus and performance in a hippocampus-dependent memory task. *Neuroscience*. (2018) 379:32–44. doi: 10.1016/j.neuroscience.2018.03.007

124. Ray B, Gaskins DL, Sajdyk TJ, Spence JP, Fitz SD, Shekhar A, et al. Restraint stress and repeated corticotrophin-releasing factor receptor activation in the amygdala both increase amyloid- β precursor protein and amyloid- β peptide but have divergent effects on brain-derived neurotrophic factor and pre-synaptic proteins in the prefrontal cortex of rats. *Neuroscience*. (2011) 184:139–50. doi: 10.1016/j.neuroscience.2011.03.067

125. Sántha P, Veszelka S, Hoyk Z, Mészáros M, Walter FR, Tóth AE, et al. Restraint stress-induced morphological changes at the blood-brain barrier in adult rats. *Front Mol Neurosci.* (2016) 8:88. doi: 10.3389/fnmol.2015.00088

126. Mongelli A, Barbi V, Zamperla MG, Atlante S, Forleo L, Nesta M, et al. Evidence for biological age acceleration and telomere shortening in COVID-19 survivors. *Int J Mol Sci.* (2021) 22:6151. doi: 10.3390/ijms22116151

127. Neurath MF, Uberla K, Ng SC. Gut as viral reservoir: lessons from gut viromes, HIV and COVID-19. *Gut.* (2021) 70:1605–8. doi: 10.1136/gutjnl-2021-324622

128. Fournier AJ, Hogan JD, Rajbhandari L, Shrestha S, Venkatesan A, Ramesh KT. Changes in Neurofilament and microtubule distribution following focal axon compression. *PLoS One.* (2015) 10:e0131617. doi: 10.1371/journal.pone.0131617

129. Gu Y, Ji Y, Zhao Y, Liu Y, Ding F, Gu X, et al. The influence of substrate stiffness on the behavior and functions of Schwann cells in culture. *Biomaterials.* (2012) 33:6672–81. doi: 10.1016/j.biomaterials.2012.06.006

130. Liptan GL. Fascia: a missing link in our understanding of the pathology of fibromyalgia. J Bodyw Mov Ther. (2010) 14:3–12. doi: 10.1016/j.jbmt.2009.08.003

131. Balfour W. Observations on the pathology and cure of rheumatism. *Edinb Med Surg J.* (1815) 11:168–87.

132. Zoppi N, Chiarelli N, Binetti S, Ritelli M, Colombi M. Dermal fibroblast-tomyofibroblast transition sustained by $\alpha v \beta 3$ integrin-ILK-Snail1/slug signaling is a common feature for hypermobile Ehlers-Danlos syndrome and hypermobility spectrum disorders. *Biochim Biophys Acta Mol basis Dis.* (2018) 1864:1010–23. doi: 10.1016/j. bbadis.2018.01.005

133. Sasabe R, Sakamoto J, Goto K, Honda Y, Kataoka H, Nakano J, et al. Effects of joint immobilization on changes in myofibroblasts and collagen in the rat knee contracture model. *J Orthop Res.* (2017) 35:1998–2006. doi: 10.1002/jor.23498

134. Squier CA. The effect of stretching on formation of myofibroblasts in mouse skin. *Cell Tissue Res.* (1981) 220:325–35. doi: 10.1007/BF00210512

135. Hinz B, Mastrangelo D, Iselin CE, Chaponnier C, Gabbiani G. Mechanical tension controls granulation tissue contractile activity and myofibroblast differentiation. *Am J Pathol.* (2001) 159:1009–20. doi: 10.1016/S0002-9440(10)61776-2

136. Ramirez-Moreno JM, Ceberino D, Gonzalez Plata A, Rebollo B, Macias Sedas P, Hariramani R, et al. Mask-associated 'de novo' headache in healthcare workers during the COVID-19 pandemic. *Occup Environ Med.* (2020) 78:548–54. doi: 10.1136/ oemed-2020-106956

137. Lim ECH, Ong BKC, Seet RCS. Headaches and the N95 face-mask amongst healthcare providers. *Acta Neurol Scand.* (2006) 113:199–202. doi: 10.1111/j.1600-0404.2005.00560.x

138. Sari E, Oztay F, Tasci AE. Vitamin D modulates E-cadherin turnover by regulating TGF-β and Wnt signalings during EMT-mediated myofibroblast differentiation in A459 cells. *J Steroid Biochem Mol Biol.* (2020) 202:105723. doi: 10.1016/j.jsbmb.2020.105723

139. Marcelin G, Silveira ALM, Martins LB, Ferreira AV, Clément K. Deciphering the cellular interplays underlying obesity-induced adipose tissue fibrosis. *J Clin Invest.* (2019) 129:4032–40. doi: 10.1172/JCI129192

140. Okifuji A, Turk DC. Sex hormones and pain in regularly menstruating women with fibromyalgia syndrome. J Pain. (2006) 7:851–9. doi: 10.1016/j.jpain.2006.04.005

141. Saber AA, Boros MJ, Mancl T, Elgamal MH, Song S, Wisadrattanapong T. The effect of laparoscopic roux-en-Y gastric bypass on fibromyalgia. *Obes Surg.* (2008) 18:652–5. doi: 10.1007/s00418-004-0667-z

142. Clocchiatti A, Ghosh S, Procopio MG, Mazzeo L, Bordignon P, Ostano P, et al. Androgen receptor functions as transcriptional repressor of cancer-associated fibroblast activation. J Clin Invest. (2018) 128:5531–48. doi: 10.1172/JCI99159

143. Fede C, Albertin G, Petrelli L, Sfriso MM, Biz C, De Caro R, et al. Hormone receptor expression in human fascial tissue. *Eur J Histochem.* (2016) 60:2710. doi: 10.4081/ejh.2016.2710

144. Jiang HS, Zhu LL, Zhang Z, Chen H, Chen Y, Dai YT. Estradiol attenuates the TGF-β1-induced conversion of primary TAFs into myofibroblasts and inhibits collagen production and myofibroblast contraction by modulating the Smad and rho/ROCK signaling pathways. *Int J Mol Med.* (2015) 36:801–7. doi: 10.3892/ijmm.2015.2288

145. Wu M, Han M, Li J, Xu X, Li T, Que L, et al. 17beta-estradiol inhibits angiotensin II-induced cardiac myofibroblast differentiation. *Eur J Pharmacol.* (2009) 616:155–9. doi: 10.1016/j.ejphar.2009.05.016

146. Tu CH, Lin CL, Yang ST, Shen WC, Chen YH. Hormonal contraceptive treatment may reduce the risk of fibromyalgia in women with dysmenorrhea: a cohort study. *J Pers Med.* (2020) 10:280. doi: 10.3390/jpm10040280

147. Blaauboer ME, Smit TH, Hanemaaijer R, Stoop R, Everts V. Cyclic mechanical stretch reduces myofibroblast differentiation of primary lung fibroblasts. *Biochem Biophys Res Commun.* (2011) 404:23–7. doi: 10.1016/j.bbrc.2010.11.033

Glossary

ACE	Angiotensin converting enzyme
CDC	Centers of Disease Control
CI	Confidence interval
COVID-19	Coronavirus disease 2019
CWP	Chronic widespread pain
CTGF	Connective tissue growth factor
DNA	Deoxyribonucleic acid
ECM	Extracellular matrix
ERK	Extracellular-signal-regulated kinase
GERD	Gastroesophageal reflux disease
GPNF	Global percutaneous needle fasciotomy
IL	Interleukin
IL-1R	Interleukin-1 receptor
IRAK-4	Interleukin-1 receptor kinase 4
JNK	JUN N-terminal kinase
МАРК	Mitogen-activated protein kinase
ME/CFS	Myalgic encephalomyelitis/chronic fatigue syndrome
MLCK	Myosin light chain kinase
NICE	National Institute for Health and Care Excellence
OR	Odds ratio
PDGF	Platelet derived growth factor
ROCK	Rho-associated protein kinase
SARS-CoV	Severe acute respiratory syndrome coronavirus
SMA	Smooth muscle actin
Smad	Small mother against decapentaplegic
SSD	Somatic symptom disorder
ssRNA	Single stranded ribonucleic acid
TAZ	Transcriptional coactivator with PDZ-binding motif
TGF-β	transforming growth factor beta
TLR	Toll-like receptor
TNF-α	tumor necrosis factor alpha
YAP-1	Yes-associated protein 1